Author's response to reviews

Title: Predictors of serum dioxin levels among adolescent boys in Chapaevsk, Russia: A cross-sectional study.

Authors:
- Russ Hauser MD, MPH, ScD (rhauser@hohp.harvard.edu)
- Paige Williams PhD (paige@jessica.harvard.edu)
- Larisa Altshul MS (laltshul@hsph.harvard.edu)
- Susan Korrick MPH (susan.korrick@channing.harvard.edu)
- Lynne Peeples MS (peeples@hsph.harvard.edu)
- Donald G Patteson Jr, PhD (dgp1@cdc.gov)
- Wayman E Turner MS (wet1@cdc.gov)
- Mary M Lee MD (LeeM@ummhc.org)
- Boris Revich MD (revich@mail.ecfor.rssi.ru)
- Oleg Sergeyev MD (endopsyh@mail.samtel.ru)

Version: 6 Date: 28 April 2005

Author's response to reviews:

We addressed the most recent suggestions. The Figure PDFs do work in several of our computers, and we suspect that the crashing problems maybe due to a particular computer memory capabilities? If the problems persist with other of your computers, please let us know.